

Format for ANSWERING REVIEWERS

November 6, 2014



Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 14143-review.doc).

Title: Nonconvulsive Status Epilepticus disguising as Hepatic Encephalopathy.

Author: Yong Min Jo, MD., Sung Wook Lee, M.D., Ph.D., Sang Young Han, M.D., Ph.D., Yang Hyun Baek, M.D., Ph.D., Ji Hye Ahn, M.D., Won Jong Choi, M.D., Ji Young Lee, M.D., Sang Ho Kim, M.D., Ph.D., Byeol A Yoon, M.D.

Name of Journal: *World Journal of Gastroenterology*

ESPS Manuscript NO: 14143

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

(1) I suggest that the therapeutic effect of nonconvulsive status epilepticus and hepatic encephalopathy in another hospital where the patient was transferred should be recorded in this case

- The final result of the patient treatment has been updated.

(2) Was it appropriate to administer treatment by mouth to a lethargic patient?

- Manuscript content has been updated.

(3) DISCUSSION It is not true that "The pathophysiology of hepatic encephalopathy remains unknown" and the discussion in the first paragraph (ref. 3) is largely inaccurate.

- Discussion's content has been corrected.

(4) My suggestion is to change the title in : Cirrhosis and nonconvulsive status epilepticus. Minor points; 1. Give normal lab values within paracenteses 2. what was the final follow-up of the patient in the other facility?

- Title has been changed.

- Manuscript content has been updated.

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

A handwritten signature in black ink, appearing to read 'Yong Min Jo', is placed over a light pink rectangular background.

Yong Min Jo, MD.

Department of Internal Medicine,

Dong-A University School of Medicine, 3 Ga 1, Dongdaesin-dong, Seo-gu, Busan, 602-715,

South Korea.

E-mail: asnow0217@naver.com

Telephone: +82-51-2405627 Fax: +82-51-2402087